Author's response to reviews

Title: Phenotypic and Genetic Characterization of a Cohort of Pediatric Wilson Disease Patients

Authors:

Tawhida Y Abdel Ghaffar (tyghaffar@gmail.com)
Solaf M Elsayed (elsayed683@yahoo.com)
Suzan Elnaghy (suzan_elnaghi@yahoo.com)
Ahmed Shadeed (Ahshadeed@hotmail.com)
Ezzat S Elsobky (elsobbyezzat@hotmail.com)
Hartmut Schmidt (hepar@ukmuenster.de)

Version: 4 Date: 27 May 2011

Author's response to reviews: see over
Date: May 27, 2011

Dear Editors,

Please find enclosed the revised manuscript titled: “Phenotypic and Genetic Characterization of a Cohort of Pediatric Wilson Disease Patients”. We thank the editors for their comments.

Major revisions:

Abstract: This should be identical in the manuscript and the submission system.

Abstract is identical in the manuscript and the submission system

Minor revisions

(Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

Removed

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows : please re-layout your table without these merged elements. Please remove the visible vertical lines from your tables.

Done

Sincerely yours,

Tawhida Y. Abdel Ghaffar, Solaf M. Elsayed, Suzan Elnaghy, Ahmed Shedeed, Ezzat S. Elsobky, Hartmut Schmidt